Author’s response to reviews

Title: Patterns of sexually transmitted infections in adolescents and youths in Dar es Salaam, Tanzania

Authors:

Guerino Chalamilla (gchalamilla@muchs-harvard.org)
Judica Mbwana (jmbwana@muchs.ac.tz)
Fred Mhalu (fmhalu@muchs.ac.tz)
Eunice Mmari (emmari@muchs.ac.tz)
Mtebe Majigo (mmajigo@muchs.ac.tz)
Andrew Swai (aswai@muchs.ac.tz)
Willy Urassa (wurassa@muchs.ac.tz)
Eric Sandstrom (eric.sandstrom@karolinska.se)

Version: 10 Date: 25 January 2006

Author’s response to reviews: see over
25\textsuperscript{th} January 2006

Dear

Thank you for accepting to publish our paper. Kindly find below two points that have been addressed in the most recent version of the manuscript.

1. Author’s contributions on page 16 has been changed to Authors’ contributions from to Author’s contributions.

2. Oriented table -2 on page 21 from landscape to portrait format.

Best Regards

Willy